Author’s response to reviews

Title: Rectal aberrant crypt foci (ACF) as a predictor of benign and malignant neoplastic lesions in the large intestine

Authors:

Marek Kowalczyk (forel@neostrada.pl)
Marcin Orłowski (m.orlowski@przychodniachelm.pl)
Łukasz Klepacki (lukasz.klepacki@uwm.edu.pl)
Krzysztof Zinkiewicz (kzinek@yahoo.com)
Waldemar Kurpiewski (wkurpiewski@interia.pl)
Dorota Kaczerska (nurdok@o2.pl)
Wiesław Pesta (wieslaw.uesta@vp.pl)
Ewa Zieliński (ewa.zielinski.bydgoszcz@wp.pl)
Piotr Siermontowski (nurdok@tlen.pl)

Version: 4 Date: 03 Jan 2020

Author’s response to reviews:

Dear Editors,

Thank you for pointing out the places to improve. As recommended, we've made all the required changes to our article:

1. In the section 'Ethics Approval and Consent to Participate' - the authors received in writing Approval for research issued by Bioethical Commission at the Faculty of Medical Sciences of the University of Warmia and Mazuria in Olsztyn

2. in the Authors Contributions section we have described in detail the individual tasks of each of the authors, using their full initials. All authors have read and approved the final manuscript.

3. All Figures and Tables have been cited in sequential order in the main manuscript text.